Author’s response to reviews

Title: Empirical treatment for tuberculosis in HIV: lessons from a cohort study of people living with HIV treated in Recife, Brazil

Authors:

Maria de Fatima PM Albuquerque (militaofatima@gmail.com)
Isabella Coimbra (isabella.coimbra@uol.com.br)
Joanna DL Batista (joannalyra@gmail.com)
Magda Maruza (magdamaruza@yahoo.com.br)
Ricardo AA Ximenes (ricardo.ximenes@pq.cnpq.br)
Heloisa L Ramos (helramos@terra.com.br)
Democrito B Miranda Filho (demofilho@uol.com.br)
Marcela L Santos (santosmlopes@gmail.com)
Laura C Rodrigues (laura.rodrigues@lshtm.ac.uk)

Version: 9 Date: 21 February 2014

Author’s response to reviews:

Dear Editor

We are pleased to send you the revised version of our manuscript, undoubtedly improved by the reviewers’ relevant suggestions. We have changed the tables and the figure 2 accordingly Valeria’s suggestion. Please, do not hesitate to contact us if you would like any further explanation.

Your sincerely,

Maria de Fátima Albuquerque